Author's response to reviews

Title: Left ventricle remodelling by double-patch sandwich technique

Authors:

Ernesto Tappainer (tap04@libero.it)
Vinicio Fiorani (fioranire@libero.it)
Nicola Pederzolli (pednik@yahoo.it)
Jacopo Manfredi (jmanfre@hotmail.it)
Andrea Nocchi (nocchi@libero.it)
Mario Zogno (mario.zogno@ospedalimantova.it)

Version: 3 Date: 8 January 2007

Author's response to reviews:

To Editor of Journal of Cardiothoracic Surgery
Mr. Vipin Zamvar

Dear Sir,
I delighted to make the required format corrections:
1) unnecessary capitals letters were removed in the title.
2) in Methods/Patients section, it was specified that some preoperative data (which are listed in Table 1) are listed and compared with the same postoperative of them (which are listed in Table 2 and 3), so that readers have to jump temporarily to Table 3 before reading Table 2.

With best regard
Mantova 8th January 2007
Dr. Ernesto Tappainer, MD